Author's response to reviews

Title: Whole-system change: case study of factors facilitating early implementation of a primary health care reform in a South African province

Authors:

Helen Schneider (hschneider@uwc.ac.za)
Rene English (Rene.English@hst.org.za)
Hanani Tabana (Hanabitabana@yahoo.com)
Thesandree Padayachee (thesandree.padayachee@hst.org.za)
Marsha Orgill (Ms.Orgill@uct.ac.za)

Version: 2
Date: 7 July 2014

Author's response to reviews:

Dear editor

I have:

added corresponding author
placed key words after abstract
added abbreviations
competing interests
placed end notes after acknowledgements
highlighted in bold all the headings